Author’s response to reviews

Title: Attenuating posttraumatic distress with omega-3 polyunsaturated fatty acids among disaster medical assistance team members after the Great East Japan Earthquake: The APOP randomized controlled trial

Authors:

Yutaka Matsuoka (yutaka@ncnp.go.jp)
Daisuke Nishi (d-nishi@umin.ac.jp)
Naoki Nakaya (nakaya-thk@umin.ac.jp)
Toshimas Sone (sone-t@tfu-mail.tfu.ac.jp)
Kei Hamazaki (keihama@med.u-toyama.ac.jp)
Tomohito Hamazaki (hamazaki@inm.u-toyama.ac.jp)
Yuichi Koido (koido@tdmc.hosp.go.jp)

Version: 2 Date: 14 July 2011

Author’s response to reviews:

July 14, 2011 (JPN)

Angelina Ilievska, MSc.
Assistant Manager
Editorial Office, BioMed Central
RE: MS 1968018135552240

Study protocol

Attenuating posttraumatic distress with omega-3 polyunsaturated fatty acids among disaster medical assistance team members after the Great East Japan Earthquake: The APOP randomized controlled trial

Yutaka Matsuoka, Daisuke Nishi, Naoki Nakaya, Toshimas Sone, Kei Hamazaki, Tomohito Hamazaki and Yuichi Koido

BMC Psychiatry

Dear Ms. Ilievska,

Thank you for your email. I send evidence of ethical approval from Ethics Committee of the National Disaster Medical Center and governmental founding (Total amount of the current year, 2011 April-2011 March: JPY 7,034,300) from Japan Science and Technology Agency for our study. I hereby submit a revised version of the manuscript with correction of current number of dead and missing after disaster, typo error, and misdescription.

I confirm that this manuscript has not been published and is not under
consideration for publication elsewhere. I have no manuscript that is under submission with other journals based on this study protocol at this time.

Yours sincerely,

Yutaka Matsuoka, MD, PhD
National Disaster Medical Center, Japan
National Center of Neurology and Psychiatry, Japan
e-mail: yutaka@ncnp.go.jp